Author's response to reviews

Title: Ion Therapy within the Trimodal Management of Superior Sulcus Tumors: The INKA trial

Authors:

Henrik Hauswald (henrik.hauswald@med.uni-heidelberg.de)
Stefan Rieken (stefan.rieken@med.uni-heidelberg.de)
Hendrik C Dienemann (hendrik.dienemann@med.uni-heidelberg.de)
Michael Thomas (michael.thomas@med.uni-heidelberg.de)
Meinhard Kieser (Meinhard.Kieser@imbi.uni-heidelberg.de)
Juergen Debus (Juergen.Debus@med.uni-heidelberg.de)
Klaus Herfarth (Klaus.Herfarth@med.uni-heidelberg.de)

Version: 2 Date: 20 November 2014

Author's response to reviews:

Dear Editor,

we reformatted the manuscript and moved the author contributions, funding and conflict of interest statement to the end of the manuscript. We added an abbreviations list. We confirm that we have forwarded the requested documentation (ethics, funding) to BMCSeriesEditorial@biomedcentral.com. However, we do not have English translations performed by the according Institutions. We would be able to translate them by our own and provide these translations.